Author’s response to reviews

Title: Randomised controlled trial of the efficacy of aerobic exercise in reducing metabolic risk in healthy older people: The Hertfordshire Physical Activity Trial

Authors:

Francis M Finucane (francis.finucane@mrc-epid.cam.ac.uk)
Jessica Horton (jh@ctu.mrc.ac.uk)
Lisa R Purslow (lisa_purslow@hotmail.com)
David B Savage (dbs23@medschl.cam.ac.uk)
Soren Brage (sb400@medschl.cam.ac.uk)
Herve Besson (hb301@medschl.cam.ac.uk)
Kenneth Horton (kgh23@medschl.cam.ac.uk)
Ema De Lucia Rolfe (ed216@medschl.cam.ac.uk)
Alison Sleigh (as626@wbic.cam.ac.uk)
Stephen J Sharp (sjs207@medschl.cam.ac.uk)
Helen Martin (hjm@mrc.soton.ac.uk)
Avan Ahie Sayer (aas@mrc.soton.ac.uk)
Cyrus Cooper (cc@mrc.soton.ac.uk)
Ulf Ekelund (ue202@medschl.cam.ac.uk)
Simon J Griffin (sjg49@medschl.cam.ac.uk)
Nicholas J Wareham (njw1004@medschl.cam.ac.uk)

Version: 3 Date: 12 June 2009

Author’s response to reviews: see over
The Editor
BMC Endocrine Disorders 12th June 2009

Re: Proof corrections for study protocol manuscript entitled “Randomised controlled trial of the efficacy of aerobic exercise in reducing metabolic risk in healthy older people: The Hertfordshire Physical Activity Trial.”

Ref: 1097774081259499 (revision)

Dear Editor,

Thank you for your recent correspondence regarding our revised manuscript. We are delighted that it has been accepted. We have incorporated each of the formatting changes requested. Please let me know if you have any further queries.

Yours sincerely,

Dr. Francis Finucane MB MD MRCPI
MRC Career Development Fellow